Author's response to reviews

Title: Efficacy and safety of a booster dose of influenza vaccination in solid organ transplant recipients: TRANSGRIPE 1-2, study protocol for a multicentre, randomized, controlled clinical trial.

Authors:

Juliana Martinez-Atienza (jumarati@gmail.com)
Clara M Rosso Fernández (claram.roso.sspa@juntadeandalucia.es)
Cristina Roca (cristinaroca85@gmail.com)
Teresa A Aydillo (teresa.aydillo@gmail.com)
Francisco Molina-Ortega (fjmolort@hotmail.com)
Ángel Bulnes (angelbulnes@hotmail.com)
Pilar Pérez-Romero (pilarper@hotmail.com)
Elisa Cordero (elisacorderom@gmail.com)
TRANSGRIPE 1-2 study Group (eecc.huvr@gmail.com)

Version: 3  
Date: 26 March 2014

Author's response to reviews: see over
Dear Dr McGavin,

As you pointed out some of the authors were removed by mistake in the documentation sent in the previous email. The actual Trial was designed in origin as a multicenter study and all the authors listed below contributed substantially in the design and the follow up work. Please find below the contribution section for each author:

JM-A: conception and design, data collection and analysis, manuscript writing and final approval of the manuscript. CMRF: conception and design, data collection and analysis, manuscript writing and final approval of the manuscript. CR: data collection and analysis, critical revision and final approval of the manuscript. TAA: data collection and analysis and final approval of the manuscript. FM-O: data collection and analysis, critical revision and final approval of the manuscript. AB-R: data collection and analysis, critical revision and final approval of the manuscript. JG: conception and design, manuscript writing and final approval of the manuscript. AM: data collection and analysis and final approval of the manuscript. JMM: conception and design, manuscript writing and final approval of the manuscript. JT: conception and design, manuscript writing and final approval of the manuscript. CF: data collection and analysis and final approval of the manuscript. JF: conception and design, manuscript writing and final approval of the manuscript. JC: conception and design, manuscript writing and final approval of the manuscript. NS: data collection and analysis and final approval of the manuscript. PM: data collection and analysis and final approval of the manuscript. MB: data collection and analysis and final approval of the manuscript. AS: data collection and analysis and final approval of the manuscript. FL-M: conception and design, manuscript writing and final approval of the manuscript. MAG: data collection and analysis and final approval of the manuscript. Miguel Angel Gómez, Ernesto Lage, Antonia Gash, MS: data collection and analysis and final approval of the manuscript. PP-R: conception and design, data collection and analysis, financial support, manuscript writing, final approval of manuscript. EC: conception and design, data collection and analysis, financial support, manuscript writing, final approval of manuscript. All authors read and approved the final manuscript.

We apologize for the inconvenience.
Thanks in advance.

Clara M. Rosso-Fernández
Clinical Pharmacologist